Author's response to reviews

Title: Duloxetine in treatment of refractory chronic tennis elbow: two case reports

Authors:

Zaid A Wani (zaidwani@gmail.com)
Shabir a dhar (shabirdhar@yahoo.co.in)
Muhammad F Bhat (mbutt72@yahoo.com)
Yasir H Rather (yasirhrather@yahoo.com)
Shano Shiekh (zaidwani@yahoo.com)

Version: 3 Date: 4 April 2008

Author's response to reviews:

• Title has been restructured.
• Additional information removed from the title page
• Abstract restructured. word count also reduced
• Information on the patient's ethnic origin in the case presentation section
• Revision of the consent statement done.
• Word count reduced
• Reference list has been reduced to 13.